Author's response to reviews

Title: Spontaneous thoracic ventral spinal subdural hematoma mimicking a tumoral lesion: a case report

Authors:

YongJian Zhu (zyjbull@gmail.com)
DeQing Peng (pengdeqing@zju.edu.cn)
Fang Shen (alfon.sf@126.com)
LinLin Wang (zjspine@zju.edu.cn)
ZhuXiao Tang (fshen@aliyun.com)
JianMin Zhang (neurosurgery@zju.edu.cn)

Version: 4
Date: 21 February 2015

Author's response to reviews: see over
Dear Editor,

We would like to submit the revised manuscript entitled "Spontaneous thoracic ventral spinal subdural hematoma mimicking a tumoral lesion: a case report" (MS: 1998874527142701), which we wish to be considered for publication as "Case Report" in your honored journal. The following are our point-to-point answers to the reviews' comments.

To reviewer 1:

1. Comments: Special detail should be put on description of the images and comparison of these images to other reported in the literature, as very well specified by one of the referees ....
   Answer: These valuable comments are greatly appreciated by the authors and revisions were made accordingly.

To Editor:

1. Comments: Please rephrase your conclusions focusing in imaging findings and the importance of including SSDH in a differential diagnosis ...
   Answer: The conclusion part of both the abstract and main manuscript were revised according to your valuable comments.

2. Comments: In addition, there are some minor language corrections that need to be done ...
   Answer: These minor corrections were made according to your comments.

3. Comments: Regarding the informed consent.....
   Answer: The patient’s personal consent was obtained and ready for your review.

Finally, I want to sincerely apologize for the delay in submitting this revised manuscript for my personal reasons. I am truly grateful to you for your kind considerations.

Thank you very much for your attention again.

Best regards,

Jian-Min Zhang, M.D.